Author’s response to reviews

Title: The Effect of Social Norm-based Intervention with Observable Behaviour on Physical Activity among Adolescents: A Randomized Controlled Trial

Authors:

Jia Jia Lee (audrey.jiajia@duke-nus.edu.sg)
Nivedita Vikas Nadkarni (nivedita.nadkarni@duke-nus.edu.sg)
Irene Teo (irene.teo@duke-nus.edu.sg)
Semra Ozdemir (semra.ozdemir@duke-nus.edu.sg)

Version: 1 Date: 21 May 2020

Author’s response to reviews:

1. Identifying Information/Consent for Publication
   - We removed housing type from Table 1.

2. Availability of Data and Materials
   - We amended this which to reflect that data will be available upon reasonable request.

Comments received on 22-May-2020

FORMATTING CHANGES:

1. Please remove 'Disclosure of potential conflicts for all named authors' in Title page.
   - Revised.

2. Please rename 'FIGURES CAPTIONS' to 'Figure legends' after references section.
   - Revised.

3. Please rename 'TABLE TITLES' to 'Table legend' after references section.
   - Revised